Author’s response to reviews

Title: Community Assessment of COPD Health Care study: a clinical audit on primary care performance variability in COPD care

Authors:
Jose Luis Lopez-Campos (lcampos@separ.es)
Maria Abad-Arranz (marieta.84@hotmail.com)
Ana Moran-Rodríguez (anamoran@comcadiz.com)
Enrique Mascarós Balaguer (enmasba@gmail.com)
Carmen Quintana Velasco (cquintanav66@gmail.com)
Laura Abad Polo (laura.abad.polo@gmail.com)
Sara Núñez Palomo (snpalomo@sociedadgrap.com)
Jaime Gonzálves Rey (gonzalvez.jaime@gmail.com)
Ana Maria Fernández Vargas (anamorronga@gmail.com)
Antonio Hidalgo Requena (ahidalgor1@yahoo.es)
Jose Manuel Helguera Quevedo (josehelg@hotmail.com)
Marina García Pardo (marinagpardo@hotmail.com)

Version: 1 Date: 17 Nov 2017

Author’s response to reviews:

Editor's comments:

1 - Please rename the ‘Introduction’ section 'Background'

RESPONSE: We have changed the wording as suggested

2 - Please revise the Declarations subsection - Ethics approval and consent to participate - of your manuscript to include the full names of the ethics committees that approved your study. We
recommend the following format: “This study was approved by the ethics committee of [xxx] Hospital [or University].” If more than five ethics committees approved your study, you may wish to list the names of the committees in an additional file.

RESPONSE: We have updated the declarations sections following the instructions provided.

3 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section.

RESPONSE: We have updated the funding section following the instructions provided.

4 - Please ensure that there are specific Authors' Contributions for all of the authors.

RESPONSE: We have included the contributions for all authors as requested.

5 - Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

RESPONSE: Yes, I can confirm that all authors have read and approved the manuscript.

6 - Group authorship: if you would like the names of the individual members of a collaboration Group to be searchable through their individual PubMed records, please ensure that the title of the collaboration Group is included on the title page and in the submission system and also include collaborating author names as the last paragraph of the “Acknowledgements” section. Please add authors in the format First Name, Middle initial(s) (optional), Last Name. You can
add institution or country information for each author if you wish, but this should be consistent across all authors.

Please note that individual names may not be present in the PubMed record at the time a published article is initially included in PubMed as it takes PubMed additional time to code this information.

RESPONSE: Thanks for giving clear instructions about this relevant aspect. We have added the information from all authors as indicated, so that all are included in PubMed records.